PEER-REVIEW REPORT

Name of journal: World Journal of Gastrointestinal Endoscopy

Manuscript NO: 76156

Title: Texture and color enhancement imaging for detecting colorectal adenomas: Good, but not good enough

SPECIFIC COMMENTS TO AUTHORS
The point that authors suggested in this article seems precise and useful in the future perspective of image enhanced endoscopy.
SPECIFIC COMMENTS TO AUTHORS

The recent paper was summarized briefly and related papers were introduced. An important paper was missing (Nishizawa et al. TXI (Texture and Color Enhancement Imaging) for Serrated Colorectal Lesions. J Clin Med. 2021 Dec 27;11(1):119). The authors should introduce this paper, too.

Re: Thank you for your comment. We added the paper based on your suggestion. Please see the highlighted sections in the letter.